RF09

TECH: Cowan
01/26/09 (Tahiti date was 01/25/09)

Pre-Flight

Extended pre-flight to change flasks. To 1630L  Opened Radome and drained from the red and purple lines. Closed them back up and taped over the holes again.  Pavel reports that the caps my be wearing out since they can be spun by hand.

Looked at VCSEL, still no laser intensity, we did shine a light up into the detector and it did respond. Suspect that Laser is either dead or misaligned. 

Started wing DSM’s and RW DSM did not boot.  Had to power on/off from power panel.
0214 block out

0222 Take off

0241 LWDSM stopped, then started again, had a quick red NOCODE. The Analog card is reporting red as is the 2DC which is not on the project. Waiting to see if the analog card comes back on it’s own.

0248 Nitrogen Dewer onboard blew the top off, there was a minor discharge of nitrogen in the cabin. Cap was secured tighter on the dewer.

0253 Rebooting LWDSM to see if the analog card will come back up.

0256 LWDSM functioning.

0307 LWDSM another problem with the analog card

19S 144W atoll pictures

0425 reset Left Dew Point due to Service Mirror Light

0505 PLWC varies with Airspeed.  This is a know issue.

0514 Resetting Left Dew Point, service mirror light on. On since 0506.

0555 Resetting Left Dew Point due to service mirror light

0558 Resetting both Dew Point because both have service mirror lights.

0559 Found that the winds were not updating.  Looked at the MPDS DSM and found that there were not any errors in the time codes but the IRIGB, both IRS’s, ADC and GPS.  This is the same issue that we have seen before. Recycled DSM at the power panel. Original drop time was 0555.

0635 Lost Satcom not noticing a redail, the service light was lite. 

0638 Left Dew Point Service Mirror light on resetting

0644 Had pilots reset Satcom Switch since the Satcom was not redialing. Had Service Available, ISDN and MPDS lights on.
0648 All Satcom lights back on and establishing communications.

0722 Left Dew Point service mirror light on resetting

0723 Right Dew Point service mirror light on resetting

0840 All DSM’s locked up, went to stopped. 

0842 Looks like we lost the server. Had a repeating Nimbus message having trouble adding data. Last good data was 083446. 

Found Power strip #3 on back of ADS rack off.  Lost power to both network hubs and the timeserver.  Since it was late in the flight we have lost the data for the rest of the flight.

0852 Landing
After landing both Dew Pointers have service mirror lights on, turned them both off.

Had to restart DSM’s and data systems started recording again.  Would have been very recoverable in flight had we had more time.

Lost data from 0834 to 0855

